\documentclass[10pt,a4paper]{article}
\usepackage[T1]{fontenc}
\usepackage{times}
\usepackage{graphicx}
\usepackage{CMG2016}

\begin{document}
\begin{thebibliography}{1}

\bibitem{bib:Lobkis2001} 
O. I. Lobkis and R. L. Weaver,
\textit{On the emergence of the Green's function in the correlations of a diffuse field.},
J. Acoust. Soc. Am. \textbf{110}, 3011-3017, (2001).

\bibitem{bib:Weaver2004} 
R. L. Weaver and O. I. Lobkis,
\textit{Diffuse fields in open systems and the emergence of Green's function.},
J. Acoust. Soc. Am. \textbf{116}, 2731-2734, (2004).

\bibitem{bib:Sanchez2006} 
F. J. {S\'{a}nchez-Sesma} and M. Campillo,
\textit{Retrieval of the Green's function from cross correlation: The canonical elastic problem.},
J. Acoust. Soc. Am. \textbf{96}, 1182-1191, (2006).

\bibitem{bib:Wapenaar2004} 
K. Wapenaar,
\textit{Retrieving the elastodynamic Green's function of an arbitrary inhomogeneous medium by cross correlation.},
Phys. Rev. Lett. \textbf{93}, 254301, (2004).

\bibitem{bib:Wapenaar2006} 
K. Wapenaar and J. Fokkema,
\textit{Green's function representations for seismic interferometry.},
Geophysics \textbf{71}, S133-S146, (2006).

\bibitem{bib:Snieder2007} 
R. Snieder,
\textit{Extracting the Green's function of attenuating heterogeneous acoustic media from uncorrelated waves.},
Geophysics \textbf{121}, 2637-2643, (2007).

\bibitem{bib:Yao2009} 
H. Yao and R.D. van der Hilst,
\textit{Analysis of ambient noise energy distribution and phase velocity bias in ambient noise tomography, with application to SE Tibet.},
Geophys. J. Int. \textbf{179}, 1113-1132, (2009).

\bibitem{bib:Harmon2010} 
N. Harmon, R. Catherine and P. Gerstoft,
\textit{Distribution of noise sources for seismic interferometry.},
Geophys. J. Int. \textbf{183}, 1470-1484, (2010).

\bibitem{bib:Tromp2010} 
J. Tromp, Y. Luo, S. Hanasoge and D. Peter,
\textit{Noise cross-correlation sensitivity kernels.},
Geophys. J. Int. \textbf{183}, 791-819, (2010).

\bibitem{bib:Fichtner2015} 
A. Fichtner,
\textit{Source-structure trade-offs in ambient noise correlations.},
Geophys. J. Int. \textbf{202}, 678-694, (2015).

%\bibitem{bib:Komatitsch2002a} 
%D. Komatitsch and J. Tromp,
%\textit{Spectral-element simulations of global seismic wave propagation, Part {I}: validation.},
%Geophys. J. Int. \textbf{149}, 390-412, (2002).
%
%\bibitem{bib:Komatitsch2002b} 
%D. Komatitsch and J. Tromp,
%\textit{Spectral-element simulations of global seismic wave propagation, Part {II}: 3-{D} models, oceans, rotation, and gravity.},
%Geophys. J. Int. \textbf{150}, 303-318, (2002).
%
%\bibitem{bib:Peter2011} 
%D. Peter, D. Komatitsch, Y. Luo, R. Martin, N. {Le Goff}, E. Casarotti, P. {Le Loher}, F. Magnoni, Q. Liu, C. Blitz, T. Nissen-Meyer, P. Basini and J. Tromp,
%\textit{Forward and adjoint simulations of seismic wave propagation on fully unstructured hexahedral meshes.},
%Geophys. J. Int. \textbf{186}, 721-739, (2011).
%
%\bibitem{bib:Fichtner2009} 
%A. Fichtner, H. Igel, H.-P. Bunge and B.L.N. Kennett,
%\textit{Simulation and inversion of seismic wave propagation on continental scales based on a spectral-element method.},
%J. Num. An. Ind. Appl. Math. \textbf{4}, 11-22, (2009).
%	
%\bibitem{bib:Gokhberg2015} 
%A. Gokhberg and A. Fichtner,
%\textit{Full-waveform inversion on heterogeneous HPC systems.},
%Computers \& Geosciences, (2015).

 \end{thebibliography}


\title{Towards a Full Waveform Ambient Noise Inversion}

\authors{
\underline{K. Sager}$^1$, L. Ermert$^1$, C. Boehm$^1$ \& A. Fichtner$^1$
}

\affiliations{
$^1$Swiss Federal Institute of Technology, Z\"{u}rich, Switzerland\\
}

\maketitle

\begin{summary}
noise tomography, full waveform inversion, source imaging, adjoint techniques.\\
\end{summary}

Noise tomography usually works under the assumption that the inter-station ambient noise correlation is equal to a scaled version of the Green function between the two receivers. This assumption, however, is only met under specific conditions, for instance, wavefield diffusivity and equipartitioning (e.g. \cite{bib:Lobkis2001}, \cite{bib:Weaver2004}, \cite{bib:Sanchez2006}) or the isotropic distribution of both mono- and dipolar uncorrelated noise sources (e.g. \cite{bib:Wapenaar2004}, \cite{bib:Wapenaar2006}, \cite{bib:Snieder2007}). These assumptions are typically not satisfied in the Earth. While attempts have been made to acknowledge heterogeneous noise source distributions (e.g. \cite{bib:Yao2009}, \cite{bib:Harmon2010}), the inconsistency between reality and theory inhibits the exploitation of the full waveform information contained in noise correlations regarding Earth structure and noise generation.\\
Realizing the need to go beyond the Green function principle and also its feasibility with current high-performance computing resources, we attempt to develop a method that consistently accounts for noise distribution, 3D heterogeneous Earth structure and the full seismic wave propagation physics. By this we intend to improve the current resolution of tomographic images of the Earth and to geographically map noise sources, thereby contributing to a better understanding of the generation of noise.\\

As an initial step towards a full waveform ambient noise inversion we develop a preliminary inversion scheme based on a 2D finite-difference code simulating correlation functions and on adjoint techniques. While the corresponding theory is mostly derived in the frequency domain (e.g. \cite{bib:Tromp2010}, \cite{bib:Fichtner2015}), time domain implementations of the seismic wave equation are widely used and further optimized in the seismology community. For future inversions using these codes, we present an efficient way to calculate kernels for the distribution of noise and for Earth structure in the time domain.\\
With respect to our final goal, a simultaneous inversion for noise distribution and Earth structure, we address the following two aspects:
\begin{enumerate} 
\item The capabilities of different misfit functionals to image wave speed anomalies and source distribution.
\item Possible source-structure trade-offs, especially to what extent unresolvable structure could be mapped into the inverted noise source distribution and vice versa.
\end{enumerate}




%%%%%%%%%%%
  


\end{document}

